Author’s response to reviews

Title: Impact of Bilateral Biopsy-Detected Prostate Cancer on an Active Surveillance Population

Authors:

Jonathan Wang (wang@UWhealth.org)
Pablo Sierra (sierra@urology.wisc.edu)
Kyle Richards (richards@urology.wisc.edu)
E. Jason Abel (abel@urology.wisc.edu)
Glenn Allen (allen@urology.wisc.edu)
Trach Downs (downs@urology.wisc.edu)
David Jarrard (jarrard@urology.wisc.edu)

Version: 2 Date: 15 Feb 2019

Author’s response to reviews:

Reviewer #1: If indeed the 2014 grading system was used to reclassify the prostate biopsies then this needs to be stated explicitly in the manuscript.

The term "Gleason score" is still found throughout the manuscript with occasional "grade group" added in parentheses. "Grade group" should be used exclusively throughout the manuscript instead of Gleason Score.

THANK YOU THESE CHANGES HAVE BEEN MADE TO THE TEXT AND TABLES.

Reviewer #2: The authors addressed reviewed comments.

1. Although the authors acknowledged in the text they used men with very low-risk disease. The same modifications were not reflected in the abstract.

THIS HAS BEEN EDITED IN THE ABSTRACT